Author’s response to reviews

Title: Simulation-based mastery learning compared to standard education for discussing diagnostic uncertainty with patients in the emergency department: a randomized controlled trial

Authors:

Danielle M McCarthy (d-mccarthy2@northwestern.edu)
Rhea Powell (rhea.powell@jefferson.edu)
Kenzie Cameron (k-cameron@northwestern.edu)
David Salzman (dsalzman005@northwestern.edu)
Dimitrios Papanagnou (dimitrios.papanagnou@jefferson.edu)
Amanda Doty (amanda.doty@jefferson.edu)
Benjamin Leiby (benjamin.leiby@jefferson.edu)
Katherine Piserchia (k-piserchia@northwestern.edu)
Matthew Klein (matthew.klein@northwestern.edu)
Xiao Zhang (XiaoChi.Zhang@jefferson.edu)
William McGaghie (wcmc@northwestern.edu)
Kristin Rising (kristin.rising@jefferson.edu)

Version: 1 Date: 06 Dec 2019

Author’s response to reviews:

Dear Editor,

We have made the requested revisions in formatting. Additionally we can clarify that the study is ongoing and that no publications containing the results of the study have been submitted or published by any journal.

Please let us know if you have any additional questions and we look forward to hearing from you.
Sincerely,

Danielle McCarthy